Author's response to reviews

Title: Script Concordance Tests: Guidelines for Construction

Authors:

Jean Paul Fournier (fournier.jp@chu-nice.fr)
Anne Demeester (Anne.Demeester@mail.ap-hm.fr)
Bernard Charlin (bernard.charlin@umontreal.ca)

Version: 6 Date: 28 April 2008

Author's response to reviews:

We put the references in the text by using consecutive numbers in brackets.

We renamed the figure section to 'Figure legends' and we place it after the reference list.

We moved tables, to text.

We cropped the figure as closely as possible to minimise white space around the image.